RESEARCH

Open Access



Ebola virus disease outbreak in Guinea: what effects on prevention of mother-tochild transmission of HIV services?

Niouma Nestor Leno^{1,2*}, Alexandre Delamou^{2,3,4}, Youssouf Koita⁵, Thierno Souleymane Diallo⁶, Abdoulaye Kaba¹, Therese Delvaux⁴, Wim Van Damme⁴ and Marie Laga⁴

Abstract

Background: An unprecedented epidemic of Ebola virus disease (EVD) affected Guinea in 2014 and 2015. It weakened the already fragile Guinean health system. This study aimed to assess the effects of the outbreak on Prevention of Mother-to-Child Transmission of HIV (PMTCT) services in 2014.

Methods: We conducted a cross-sectional retrospective study. Data was collected from 60 public health centers (30 in the EVD affected areas and 30 in the unaffected areas). The comparison of PMTCT indicators between the period before Ebola (2013) and during Ebola (2014) was done using the t- test for the means and the Chi-square test for the proportions.

Results: This study showed a substantial and significant reduction in the mean number of antenatal care visits (ANC) in the affected localities, 1617 ± 53 in 2013 versus 1065 ± 29 in 2014, p = 0.0004. This would represent 41% drop in health facilities' performance. On the other hand, in the unaffected localities, the fall was not significant. The same observations were made about the number of HIV tests performed for pregnant women and the number of HIV positive pregnant women initiating ARVs. The study also noted an increase in the proportion of women tested HIV+ but who did not receive ARVs (12% in 2013 versus 44% in 2014) and HIV+ pregnant women who delivered at home (18% in 2014 versus 7% in 2013).

Conclusion: This study showed that PMTCT services, which are one of the key services to improve maternal and child health, were affected in Guinea during this Ebola outbreak in 2014 compared to 2013.

Keywords: Ebola, Effects, PMTCT, Health system, Guinea

Plain English Summary

Why was this study conducted?

This study aimed to assess the effects of the outbreak on PMTCT services in Guinée, in 2014.

How was the study conducted?

We assessed changes in PMTCT indicators during the first year (2014) of the epidemic. We have included 60 public health centers, including 30 in affected areas and 30 in unaffected areas. The data collected covers the

* Correspondence: nnleno81@gmail.com

Full list of author information is available at the end of the article

period from January 1, 2013 to December 31, 2014. The ANC and PMTCT registers were used for this collection. For each of the two areas, we compared the 2013 values to the 2014 values to estimate the changes that occurred as effect of the outbreak.

What was found in this study?

In 2014, the number of pregnant women attending ANC visits fell by 41% in the EVD affected localities, compared to only 7% in the unaffected localities.

All PMTCT indicators fell sharply during the 2014 epidemic compared with 2013. This decline was statistically more pronounced in Ebola-affected areas than in non-affected areas.



© The Author(s). 2018 **Open Access** This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated.

¹Bureau de Stratégie et de Développement du Ministère de la Santé, Conakry, Guinea

²Chair de Santé Publique de l'Université Gamal Abdel Nasser de Conakry, Conakry, Guinea

What have we learned?

The weak capacity of the health system to respond to emergencies while continuing the provision of routine services and the mistrust between health care workers and communities contributed to low utilization of health services during this epidemic. It's important to continue strengthening the Guinean health system to make it more resilient to mitigate the long-term effects of the epidemic and prepare it for future health crises. To this end, special emphasis should be placed on human resources for health.

Article summary

Strengths

- The comparison of indicators before and during the epidemic, as well as in Ebola-affected and unaffected areas, allow for a better estimate of the effect of the epidemic on PMTCT services;
- The study included PMTCT centers that did not have interruptions in supply (and therefore services) on a monthly basis of 5 or more days in 2013 to account for biases related to the unavailability of supplies.

Limitation

- The use of routine data could include underreporting bias because of the lack of regular monitoring of management tools.
- The population coverage could not be calculated because denominator data were no available. This is because the estimate of the "Spectrum" software used in Guinea could not provide disaggregated target values by region, district, or heath facilities;
- Lack of information on how the pregnant women perceived the EDV epidemic situation and on the outcome of exposed children lost to follow-up.
- Retrospective data collection (design used in this study) could be one of the limitations of this study due to biases related to the management of data sources (e.g. registry) and the problem of reporting (under or overreporting).

Background

Introduction

An unprecedented epidemic of the Ebola virus disease (EVD) affected mainly three countries in West Africa, including Guinea, Sierra Leone and Liberia in 2014 and 2015. As of 17 January 2016, the cumulative number of cases in these three countries was 28,638, of which 11,316 died [1], representing a 39.51% lethality. This epidemic remains the largest in the world in terms of the number of cases, deaths and geographical distribution since the causative virus was isolated in 1970 [2].

In Guinea, the EVD outbreak experienced three major waves of transmission in 2014. The first two were located in Conakry and some prefectures of the forest region and the third was the phase of intense transmission at national level [3].

As of December 29, 2015, 3804 Ebola cases were reported in Guinea. Of these, there were 2536 deaths, a fatality rate of 66.6% [4]. In addition, it should be noted that more than 70% of country's total Ebola cases occurred in 2014. Also, the highest number of human-to-human transmissions of the Ebola virus occurred in 2014 compared to 2015 [5]. 148 Ebola cases were confirmed among Health Care Workers as of 28 December 2014, with a fatality rate of 58.78% [5].

These alarming statistics suggest that the Ebola epidemic had probably significant indirect effects beyond the mortalities and morbidities it caused in the social and development sectors of affected communities between 2014 and 2015 [3]. The scale of the epidemic further affected the fragile health systems in the three most affected countries (Guinea, Liberia and Sierra Leone), in a way that vital and limited resources had to be reallocated to cope with the crisis, reducing the ability of countries to manage health problems [2, 6]. One aspect of the health system that could be hardly hit was the provision of basic health care services [3].

Plucinski et al., in their study conducted in 2014 noted a steady decrease of 11% in the use of health services for malaria at the operational level in Guinea during the Ebola period (2014) compared to the year before the outbreak (2013). They also noted a 15% decrease in the number of malaria cases reported in these structures [3].

It should also be noted that given the high maternal and infant mortality ratio that existed well before Ebola [7, 8], disruption of health services could be particularly damaging to maternal and child health services. A study conducted by Camara et al. [9] in 2016 showed a 51% decrease in the use of family planning services during the Ebola period in Macenta health district (Guinea) compared to the period before Ebola. The same study reported a significant 41% decrease in the number of women visiting health facilities for antenatal care [9]. Because PMTCT services are integrated into the minimum package of primary health care in Guinea, it would be expected that these services would be similarly affected by the outbreak.

Prevention of vertical HIV transmission is one of the strategies to eradicate HIV worldwide by 2030 [10, 11]; and one of the priorities of Guinea's national AIDS strategy [12]. Many activities under this strategy occur in health care settings. Therefore, any reduction in the use of these services [13, 14], due to the EVD outbreak, could negatively impact the achievement of national targets for reducing vertical transmission of HIV and de facto reducing maternal mortality and child care in Guinea.

The purpose of this study was to describe and analyze the effects of the EVD outbreak in 2014 on PMTCT service provision in Guinea.

Country general context

Guinea is located in West Africa with a population estimated at 11.7 million inhabitants in 2014, of whom 52% are women [15] and more than half (65%) of this population lives in rural areas. The country is one of the poorest countries in the world with 55.2% of the population living in poverty [7]. The health status of the population remains worrying, with high rates of maternal and infant mortality and high prevalence of certain diseases such as malaria, and adding to these are emerging and re-emerging diseases such as Ebola (2013–2015). Maternal mortality ratio increased from 980 to 724 per 100,000 live births between 2005 and 2012 [7].

In the interim report of 2016 multiple indicators clusters survey (MICS 2016) found a maternal mortality rate of 550 per 100,000 live births in Guinea, suggesting that Guinea remains among the countries with high maternal mortality rate in Africa [8]. In addition, the health status of children under 5 remains a concern. Children in this age group are among the most affected by malaria, which is the leading cause of morbidity and mortality in Guinea [7, 8]. In this age group (children under 5 years), the mortality rate before the first birthday is 88 per 1000 live births [8]. This situation could be explained by the weakness of Guinea's traditional health care system, which is based on primary health care [16]. The system currently faces many challenges, including a low coverage of human resources for health (0.5 health personnel per 1000 inhabitants in 2016 for a norm of 2.3 per 1000 inhabitants), low immunization status in children (only 26% of children aged 12-23 months are fully immunized) and a concurrent high out of pocket payment for health expenditure (62% in 2014) [16].

HIV and PMTCT care in Guinea

In Guinea, the HIV epidemic is of generalized type with an estimated national prevalence of 1.7% among people aged 15 to 49 years in 2012 [7]. According to recent estimates from UNAIDS, about 120,000 people were living with HIV in Guinea in 2014 4600 people died as a result of this disease [17].

The HIV prevalence is higher among women (2.1%) than men (1.2%) of the same age group (15–49 years) [7]. This disparity is more pronounced among pregnant women aged 15–49, whose HIV prevalence was 2.5% in 2008 and 3.56% in 2015 [18, 19]. According to estimates of the national AIDS control program, in 2014, 14% of new infections occurred among children, with 64% of

them relating to vertical transmission (mother-child) [12]. The country's PMTCT program was initiated in 2003 with seven pilot sites, including five in Conakry and two in Forécariah. A Scaling-up Plan was developed in 2008. A National Plan to Eliminate Mother-to-Child Transmission of HIV (e-PMTCT) was also developed in 2012 [20]. The current treatment regimen, in the context of PMTCT, is based on triple therapy known as "Option B+", as recommended by the World Health Organization [21, 22]. It is initiated immediately after the diagnosis of HIV infection and continues for the rest of the mother's life. Strategies identified by the national AIDS program to achieve the goals of eliminating mother-to-child transmission of HIV in Guinea by 2017 include lifelong administration of antiretroviral to HIVpositive pregnant women (Option B+), decentralization of PMTCT services to peripheral health facilities, use of community health workers to engage target populations at PMTCT sites, active search for those lost to followup, and use of blotting papers to collect and transport blood samples for the virological diagnosis of HIV [12]. These activities take place in health centers providing ANC services. The geographical coverage of public facilities that have integrated PMTCT activities remains low (65% of the existing 453 facilities in 2016) [23]. In 2013, there were already significant PMTCT gaps to be addressed in Guinea before the EVD outbreak occurs in 2014. The epidemic might have therefore aggravated these already existing gaps.

Methods

Study's type

This was a retrospective cross-sectional evaluative study using routine data from public health facilities.

Study population

The study targeted three categories of users of maternal and child health services including pregnant women, HIV-positive pregnant women and children born to HIV-positive mothers enrolled in public health centers providing prenatal and PMTCT services from 1 January 2013 to 31 December 2014.

We chosed 2014 as the year of comparison because the epidemic was not yet widespread in the country compared to 2015 and thus, could allow good comparison of different geographical areas.

The study included a representative sample of health facilities from across the country. Due to the wide disparity in the reporting of EVD cases, we divided the country into two strata. Stratum 1 consisted of the five most EVD affected health districts in 2014 and included Gueckédou, Macenta, Nzérékoré, Coyah and Conakry. Stratum 2 consisted of five randomly selected health districts among the districts not affected by the EVD in 2014. It included the health districts of Labe, Mamou, Boké, Kankan and Siguiri. This stratification was done to allow for comparisons, not only between the periods before (2013) and during (2014) the EVD outbreak in the same area, but also among the EVD affected and non-affected localities. Within each of the two strata, a random selection of 30 health centers offering both antenatal care and PMTCT services was carried out [24].

The selection of the study sites was conducted in steps. Step 1 included the establishment of the exhaustive list of health centre's offering both ANC and PMTCT services by stratum. Step 2 included the identification of ANC/PMTCT sites that did not experience PMTCT services disruption for more than 5 days per month in 2013. Step 3 included the random selection of 30 ANC/PMTCT sites in each of the two strata, using the list established in step 2.

Computer random generated numbers were used for the different selections. When a selected site did not meet the selection criteria, it was replaced by the one immediately following in the sampling frame, and so on.

Data collection

Antenatal care (ANC) and Prevention of Mother-to-Child Transmission of HIV (PMTCT) registers of the included public health facilities constituted the source for data collection.

A pre-elaborated survey questionnaire was used for data collection. The data collectors were selected from the national STI/HIV/AIDS prevention and management program (Programme National de Prise en Charge Sanitaire et Prévention des infections Sexuellement Transmissible et le Virus de l'Immunodéficience Humaine, (PNPCSP/IST/VIH) and the National AIDS Control Committee (Comité National de Lutte contre le Sida, CNLS) in Guinea.

They were selected based on their experience and knowledge of interventions to prevent mother-to-child transmission. Data collection was collected retrospectively. The approach was motivated by the low completeness of the statistical reports submitted to the national health information system and the results of previous evaluations showing discrepancies between data reported by health facilities and the actual data captured from facility registers [25]. The study focused on the PMTCT cascade. To this end, the following data was collected: 1) the number of pregnant women receiving ANC, representing pregnant women who made their first ANC visit for the current pregnancy and who were registered in the CPN registers; 2) the number of pregnant women tested for HIV which includes pregnant women who made their first ANC visit and received counselling about HIV testing, were screened for HIV and were registered in PMTCT registers; 3) the number of HIV+

pregnant women, representing women seen for their first ANC visit who were tested HIV positive and were registered in the PMTCT registers; 4) the number of HIV-positive pregnant women on ARVs, including HIVpositive pregnant women who initiated ARV treatment in PMTCT services and who were registered in PMTCT registers; 5) the number of exposed children on ARVs, representing infants born to HIV-positive mothers receiving antiretroviral prophylaxis for PMTCT; 6) the number of HIV+ pregnant women who gave birth at home, representing HIV-positive pregnant women registered in health facilities for PMTCT monitoring but who gave birth outside a health facility.

Data analysis

Data was analyzed using the EPI Info 3.5.4 software (CDC Atlanta, USA). The data of this study focused on the overall performance of health centers offering prevention of mother-to-child transmission services. After entering all the data into an excel database, we performed a quality check as follows: (i) a complete check of all the values of the study variables in the database to check for missing values and outliers and make necessary corrections and, (ii) quality check of a sample of the data collection cards to make comparisons between the database and paper copies.

The comparison covered two periods: period 1 from January 1 to December 31, 2013 (Pre-Ebola period) and period 2 from January 1 to December 31, 2014 (the first year of the Ebola epidemic in Guinea).

We used mean and standard deviations (SD) to examine the differences between absolute values. The t-test was used for normal distributions and the Wilcoxon test for asymmetric distributions. The absolute difference in performance between 2014 and 2013 was calculated for each stratum. We used the Chi-square test (χ 2) to compare proportions. The difference was considered statistically significant for values of p < 0.05.

Three main proportions were calculated including (i) the proportion of HIV-positive pregnant women on ARVs, obtained by dividing the number of HIV-positive pregnant women who initiated ARV treatment during our study period by the number of all pregnant women tested HIV-positive during our study period; (ii) the proportion of exposed children on ARVs, obtained by dividing the number of children born to HIV-positive mothers who received ARV prophylaxis by the number of pregnant women tested positive during our study period; and (iii) the proportion of HIV-infected pregnant women giving birth at home, obtained by dividing the number of HIV-positive pregnant women tested positive during the number of pregnant women who gave birth at home by the number of pregnant women tested positive during the study period.

These proportions are based solely on the performance of the sites and not on the coverage of the population. This is because of the lack of available data that could provide disaggregated target values by region, district, or care structure. The estimate of the "Spectrum" software used in Guinea only gives nationwide estimates and projections.

Results

A total of 60 public health centers offering both ANC and PMTCT services were included in the analysis.

Among facilities located in EVD affected areas, there was a significant reduction in the mean number of pregnant women attending ANC visits, 1617 ± 53 in 2013 compared to 1065 ± 29 in 2014 (p = 0.0004. In non EVD affected localities, however, the decrease was not statistically significant (from 1817 ± 331 in 2013 to 1689 ± 280 in 2014 with, p = 0.5696 (Table 1). The same findings were made for the number of pregnant women tested for HIV, the number of HIV-positive pregnant women on ARVs and the number of exposed children who received prophylactic ARVs (Table 1).

Figure 1a presents the monthly trend of PMTCT activities in EVD affected localities in 2014. It appears that the monthly number of pregnant women attending ANC visits was more or less stable in 2013 with a peak in July. However, in 2014, the trend was modified. The fall began between February and March, peaking between July and September 2014 with a slight increase between October and December 2014 (Fig. 1a). In contrast, in unaffected localities, the monthly trends of the mean number of ANCs visits in 2013 and 2014 are almost similar (Fig. 1b).

Differences in the performance of health facilities between 2013 and 2014 in terms of PMTCT services are presented in Table 2. In 2014, the number of pregnant women attending ANC visits fell by 41% in the EVD affected localities, compared to only 7% in the unaffected localities. The fall in the number of HIV+ pregnant women who received ARVs was 55% in the EVD affected area compared to only 8% in the unaffected area. Similar trends were observed for two other indicators including the number of pregnant women tested for HIV (with a 53% drop) and the number of exposed children receiving prophylactic ARVs (with a 69% drop) (Table 2). (In EVD affected areas, the changes were greater during the third quarter of 2014 compared to the same period in 2013. Regarding the number of HIV-infected pregnant women on ARVs, the fall did not change between the different quarters of 2014, except for the first quarter (Table 2).

Figure 2a compares the proportion of pregnant women tested positive for HIV who did not receive ARVs in order to reduce the risk of mother-to-child transmission of HIV, but also to improve their own health status. This proportion increased during the first year of the Ebola outbreak in the EVD affected areas, from 12% in 2013 to 44% in 2014, p < 0.05. However, in the unaffected areas, there were no statistically significant changes (12% in 2013 compared to 13% in 2014, p = 0.427) (Fig. 2b). Figure 3 shows that the proportion of HIV+ pregnant women who gave birth at home increased significantly during the EVD outbreak (7% in 2013 versus 18% in 2014) in the affected areas, P < 0.0001.

Discussion

Our study shows that in Guinea, the use of PMTCT services significantly decreased during the EVD outbreak (2014) compared to the period before the EVD (2013). In addition, our results show that the decrease in the use of PMTCT services was significantly more pronounced in the EVD affected areas compared to the areas not affected by the EVD in 2014.

Our results corroborate those reported by Delamou et al., in a study on the effects of the EVD on maternal and child health services in the forest region of Guinea [26]. The authors noted a significant reduction in the number of ANC visits and institutional deliveries during the epidemic compared to the pre-epidemic period. Similarly, they reported a significant drop in the number of immunized children and expressed concerns about the postepidemic trends that were still below figures of the preepidemic period. Despite the methodological differences between our two studies (completeness of data in their

 Table 1 PMTCT indicators comparison by year and by Ebola areas, 2014, Guinea

INDICATORS	Ebola areas affe means±SD	ected		Ebola unaffected areas means±SD			
	2013	2014	P-value	2013	2014	P-value	
Pregnant women receiving ANC	1617 ± 53	1065 ± 29	0.0004	1817 ± 331	1689 ± 280	0.5696	
Pregnant women tested for HIV	1460 ± 266	717 ± 140	0.0000	1622 ± 247	1379 ± 212	0.1556	
Pregnant women HIV+	44 ± 13	31 ± 9	0.1292	27 ± 7	27 ± 6	0.9723	
HIV-positive pregnant women on ARVs	28 ± 6	13 ± 3	0.0000*	24 ± 7	22 ± 5	0.6513	
Exposed children receiving prophylactic ARVs	22 ± 5	6 ± 1	0.0000*	17 ± 5	16 ± 3	0.6020	

*=no - parametric test

Number of sites: 30 in the Ebola zone and 30 in the non-Ebola zone



study and purposive sampling in ours), we reached the same conclusions. At the same time, this could also mean that PMTCT services would not probably regain the levels they had before the epidemic. Our results are also similar to the findings of a review of 22 studies conducted in 2016 by Ribacke et al., which reported a decrease in the number of ANC visits by 27% during the last six months of 2014 at national level in Sierra Leone and by 50% in Moyamba health district [13].

The significant drop in the number of ANC visits, pregnant women tested for HIV, and the number of HIV-positive pregnant women receiving antiretroviral

therapy may be due to a several factors. The first is the fear of communities to contract EVD in health facilities where many health workers were already infected [24]. This fear resulted from mistrust between health service providers and the community ([26]. This was the consequence of rumors around Ebola. One of the rumors circulating said that there was acomplicity between the political authorities and the International Non-Governmental Organizations (NGOs) to invent this epidemic in order to attract external funding and make financial profits. The second factor, a direct consequence of contaminations (lack of knowledge of the transmission modes and low protection of health workers) and deaths

Table 2 Changing in indicators by quarter and by the EVD status of the study areas, Guinea

	Area affected by Ebola				Area not - affected by Ebola					
	Q1	Q2	Q3	Q4	Global	Q1	Q2	Q3	Q4	Global
Pregnant women receiving ANC	-9%	-47%	- 67%	-46%	-41%	6%	-5%	-16%	-13%	-7%
Pregnant women tested for HIV	-8%	-69%	- 77%	-62%	- 53%	11%	- 14%	-31%	-25%	- 15%
Pregnant women HIV+	-15%	-19%	-29%	-52%	-28%	23%	16%	-18%	- 19%	-1%
HIV-positive pregnant women on ARVs	-14%	- 71%	-68%	-77%	-55%	22%	0%	-27%	-23%	-8%
Exposed children receiving prophylactic ARVs	-28%	-86%	- 86%	-89%	-69%	25%	-16%	-26%	-15%	-9%

Q Quarter, Q1 January to March, Q2 April to June, Q3 July to September, Q4 October to December



recorded among health personnel, would be the desertion of health facilities by the staff [27]. As a result, the supply and quality of health services might have been reduced. The third factor is the temporary closure of some health facilities during the peak of the epidemic [3]. The fourth factor is about the reorientation of a large number of health workers and available resources to EVD response activities (epidemiological surveillance, contact tracing, awareness raising) deemed less risky and more financially profitable.

The reduction in the number of ANC visits might have resulted in reduced demand for HIV testing among pregnant women in health centers. In fact, some women,



even those who attended ANC, have been able to refuse HIV testing because of rumors about EVD transmission by blood and needles. Even though this information has not been collected, it is possible that this situation has led to an increase in the rate of vertical transmission of HIV due to ignorance of the serological status of pregnant women and increased viral load in women living with HIV, known but poorly managed.

Leuenberger et al. [26] also observed a 46% decrease in the number of patients screened for HIV in 2014 in the Medical Center of the African Phil Mission in Macenta (one of the most EVD affected district in forest Guinea in 2014). In addition, a report by UNICEF on the impact of EVD in Sierra Leone mentioned that the use of PMTCT services in the country fell by 23% in 2014 compared to 2013 [28].

EVD has also disrupted the HIV supply and distribution system at the national level by reducing the access of pregnant women to ARVs. EVD occurred in a context where HIV product management was still centralized and where international technical assistance may have been lacking due to travel restrictions and border closures [13].

Our study also showed a significant increase in the proportion of HIV-positive pregnant women who gave birth at home and the high rate of loss to follow-up among exposed children. This situation could be explained by two factors. First, even after having followed the ANC visit, women would fear to be contaminated by the EVD during deliveries in health facilities that were stigmatized for Ebola. Second, the reorientation of many community mediators, usually in charge of follow-up, to the EVD response activities might have negatively affected PMTCT services outcomes. A high number of HIV patients lost to follow-up (18.4% at six months) was reported in the study by Leuenberger in Macenta district [29]. Similar figures were reported in the systematic review by Ribacke et al., which showed a significant increase (p < 0.001) in the number of HIV patients lost to follow-up at 3 months, between June and October 2014 in Liberia [13]. This could have led to an increased risk of mother-to-child transmission of HIV (MTCT) due to increased numbers of unsafe home deliveries, jeopardizing the investments of countries and their partners in the reduction of maternal and infant mortality that were in place before the EVD.

Our study had some limitations, in particular the lack of information on the perception of pregnant women and the future of exposed children lost to follow-up. In addition, only sites offering both ANC and PMTCT services were included in the study. This does not provide information about those offering only ANC. In addition, the collection of routine data could include underreporting bias because of the lack of regular monitoring of management tools as health care workers were occupied with the EVD response. Furthermore, the population coverage was not calculated because of the lack of target populations for study areas. Nevertheless, our study is one of the few studies that compared indicators before and during the EVD and compared indicators between EVD affected and non-affected areas in 2014. In addition, the study included PMTCT sites that did not experience stock rupture of more than five days a month in 2013 to account for biases related to unavailability of supplies.

Conclusion

The present study estimated the collateral effects of the EVD outbreak on PMTCT services in Guinea. All PMTCT indicators experienced a drastic fall during the 2014 epidemic compared to 2013. This fall was more pronounced in EVD-affected areas compared to nonaffected ones. Lost to follow-up and home deliveries (unsecured) increased significantly. The weak capacity of the health system to respond to emergencies while continuing the provision of routine services (resilience) and the mistrust between health care workers and communities contributed to low utilization of health services during this epidemic. It is important to continue strengthening the Guinean health system to make it more resilient to mitigate the long-term effects of the epidemic and prepare it for future health crises. To this end, special emphasis should be placed on human resources health.

Abbreviations

ANC: Antenatal Care; ARVs: Antiretroviral; e-PMTCT: Eliminate Mother-to-Child Transmission of HIV; EVD: Ebola Virus Disease; HIV: Human Immunodeficiency Virus; IST: Infection Sexuellement Transmissible; Option B+: Antiretroviral therapy for all pregnant women tested positive for any immune status for life; PMTCT: Prevention of Mother-to-Child Transmission of HIV; PNCSP: Programme National de Prise en Charge Sanitaire et de Prévention des IST/VIH/Sida; SECNLS: Secrétariat Exécutif du Comité National de Lutte contre le Sida; Sida: Syndrome de l'Immunodéficience Acquise; UNICEF: United Nations Children's Fund; VIH: Virus de l'Immunodéficience Humaine

Acknowledgments

Our thanks go to the actors of the national STI/HIV/AIDS prevention and management program who worked hard during this epidemic and who facilitated the collection of the data used in this paper. We also thank the researchers from the Department of Public Health of the Institute of Tropical Medicine (ITM) of Antwerp, Belgium for their technical support in the design, data analysis and writing of this paper.

Funding

This study was conducted as part of the implementation of the Monitoring and Evaluation and Research activities of the national AIDS control program in Guinea. The supervision of HIV delivery services that enabled the collection of data from this study was funded by the National Development Budget (BND) of Guinea and the Global Fund to Fight AIDS Tuberculosis and Malaria on behalf of the new funding model's grant 2013–2017. This grant is co-managed by two Principal Recipients: the National AIDS Control Committee (CNLS) of Guinea and the NGO "Population Service International" (PSI). PNPCSP funds, one of the sub-recipients of this grant, were used for the implementation of data collection. The opinions expressed herein do not necessarily reflect those of the NCSPP, the Global Fund or CNLS.

Data sharing

The databases used in this article are available in Excel format from the authors. They can be made available to a requestor after agreement of the authors.

Authors' contributions

LNN, LM initiated the study and developed the study protocol. All authors approved the study protocol. KY, TSD and NNL collected data. NNL, AD, TD and LM performed the analysis and wrote the manuscript. All authors commented on the manuscript project, then read and approved the final manuscript.

Ethics approval and consent to participate

The study received administrative authorization from the Ministry of Health of Guinea. All the data collected were anonymous and accessible only by the staff involved in the study.

Consent for publication

The protocol of this study doesn't include data from an individual person; therefore, consent to publication is not applicable.

Competing interests

The authors declare that they have no competing interests.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Author details

¹Bureau de Stratégie et de Développement du Ministère de la Santé, Conakry, Guinea. ²Chair de Santé Publique de l'Université Gamal Abdel Nasser de Conakry, Conakry, Guinea. ³Centre National de Formation et de Recherche en Santé Rurale de Maferinyah, Forécariah, Guinea. ⁴Department of Public Health, Institute of Tropical Medicine, Antwerpen, Belgium. ⁵Programme National de Prise en Charge Sanitaire et de Prévention des IST/ VIH/Sida (PNPCSP) du Ministère de la Santé, Conakry, Guinea. ⁶Secrétariat Exécutif du Comité National de Lutte contre le Sida (SECNLS), Conakry, Guinea.

Received: 8 December 2017 Accepted: 2 April 2018 Published online: 10 April 2018

References

- Organisation Mondiale de la Sante. Rapport de situation sur la flambée de maladie à virus Ebola. 2016;(869). http://apps.who.int/iris/bitstream/handle/ 10665/204272/ebolasitrep_20Jan2016_fre.pdf?sequence=1.
- Kieny M-P. Ebola and health systems: now is the time for change [internet]: Media center, WHO. Available from: http://who.int/mediacentre/ commentaries/health-systems-ebola/en/
- Plucinski MM, Guilavogui T, Sidikiba S, Diakité N, Diakité S, Dioubaté M, et al. Effect of the Ebola-virus-disease epidemic on malaria case management in Guinea, 2014: a cross-sectional survey of health facilities. Lancet Infect Dis. 2015;15(9):1017–23.
- OMS. Rapport de situation sur la flambée de Maladie à virus Ebola au 29 décembre 2015. 2016.
- 5. Organisation Mondiale de la Sante. Feuille de route pour la riposte au virus Ebola, Rapport de situation au 31 décembre 2014; 2015. p. 1–20.
- 6. Save the Children. Épidémie Ebola: D'abord une crise Sanitaire; 2014. p. 2–5.
- Institut National des Statistiques de Guinée du Ministère du Plan. Enquête Démographique et de Santé et à Indicateurs Multiples, Guinée 2012. Claverton: ellNS & Macro. p. 2012.
- Ministere du Plan et de la Coopération internationale/Institut de Statistique. Guinée_Rapport final des résultats cles. Enquête par grappes à indicateurs multiples 2016. 2017.
- Camara BS, Delamou A, Diro E, Béavogui AH, El Ayadi AM, Sidibé S, et al. Effect of the 2014/2015 Ebola outbreak on reproductive health services in a rural district of Guinea: an ecological study. Trans R Soc Trop Med Hyg 2017;111(1):22–29.
- UNAIDS. Accélérer la riposte. Mettre fin à l'épidémie de Sida d'ici à 2030; 2014. p. 19–20.

- 11. Ambia J, Mandala J. A systematic review of interventions to improve prevention of mother-to-child HIV transmission service delivery and promote retention. J Int AIDS Soc. 2016;19(1):1–11.
- 12. Comite National de Lutte contre le Sida de Guinée. Cadre Stratégique National de lutte Contre les IST/VIH/sida pour la periode 2013-2017de Guinée. 2013.
- Brolin Ribacke KJ, Saulnier DD, Eriksson A, von Schreeb J. Effects of the West Africa ebola virus disease on health-care utilization - a systematic review. Front Public Health. 2016;4:222.
- Parpia AS, Ndeffo-mbah ML, Wenzel NS, Galvani AP. Effects of Response to 2014–2015 Ebola Outbreak on Deaths from Malaria, HIV / AIDS, and Tuberculosis, West Africa. Emerg Infect Dis. 2016;22(3):433–41.
- Institut National des Statistiques de Guinée du Ministère du Plan. Enquête Légère pour l'Evaluation de la Pauvreté en Guinée. 2002.
- Ministère de la Santé G. Plan National de Développement Sanitaire 2015-2024. p. 2015.
- 17. UNAIDS. Zimbabwe HIV and AIDS estimates [Internet]. 2015. Available from: http://www.unaids.org/en/regionscountries/countries/guinea/
- Secrétariat Exécutif du Comité National de Lutte contre le Sida-Guinée. Enquête Nationale de Surveillance sentinelle du VIH chez les Femmes enceintes en Guinée. Appui financier du Fonds mondial. 2015.
- Ministere de la Sante et l'Hygiene Publique. Enquête Nationale de Surveillance sentinelle du VIH chez les Femmes enceintes en Guinée. Appui technique de GIZ et SE/CNLS. 2008.
- 20. PNPCSP/MS-Guinée. Plan National d'élimination de la transmission du VIH de la mère à l'enfant en Guinée (eTME). 2012.
- WHO (World Health Organization). Summary: Consolidated guidelines on the use of ARV drugs for treating and preventing HIV infection - Key features and recommendations 2013; 2013. p. 16.
- World Health Organisation. Consolidated guidelines on the use of antiretroviral drugs for treating and preventing HIV infection: recommendations for a public health approach - 2nd ed. WHO Guidel. 2016;(June):480.
- SECNLS/PNPSCP/PSI. Cartagraphie nationale des services VIH (PEC, PTME, CDV) et des sites de fréquentation des populations clés (HSH et PS) en Guinée. RAPPORT FINAL. 2017.
- 24. Institut de Médecine tropicale d'Anvers (IMT). Belgique Epidemiologie et statistique de base; 2014. p. 1–158.
- AMP-Agence de Médecine Préventive de Paris. Evaluation de la gestion et de la qualité des données du Programme Elargi de Vaccination (PEV) de la Guinée. 2014.
- 26. Alexandre D. Effect of Ebola virus disease on maternal and child health services in Guinea : A retrospective observational cohort study Eff ect of Ebola virus disease on maternal and child health services in Guinea : a retrospective observational cohort study. Lancet. The author(s). Published by Elsevier Itd. This is an Open Access article under the CC BY-NC-ND license; 2017;(February).
- Ribacke KJB, Duinen AJ Van, Nordenstedt H, Molnes R, Froseth TW, Darj E, et al. The impact of the West Africa Ebola outbreak on obstetric health care in Sierra 2016;1–12.
- Unicef. Sierra Leone Health Facility Survey 2014: Assessing the impact of the EVD outbreak on health Systems in Sierra Leone. Freetown: UNICEF. p. 2015. https://www.unicef.org/emergencies/ebola/files/SL_Health_Facility_Survey_ 2014Dec3.pdf.
- Leuenberger D, Hebelamou J, Strahm S, De Rekeneire N, Balestre E, Wandeler G, Dabis F, IeDEA West Africa study group. Impact of the Ebola epidemic on general and HIV care in Macenta, Forest Guinea, 2014. AIDS. 2015;29(14):1883-7.